Title: Missense mutations in Desmocollin-2 N-terminus, associated with arrhythmogenic right ventricular cardiomyopathy, affect intracellular localization of desmocollin-2 in vitro

Authors:

Giorgia Beffagna (giorgia.beffagna@unipd.it)
Marzia De Bortoli (marzia.debortoli@unipd.it)
Andrea Nava (andrea.nava@unipd.it)
Michela Salamon (mik@cribi.unipd.it)
Alessandra Lorenzon (alessandra.lorenzon@unipd.it)
Manuela Zaccurlo (manuela.zaccurlo@unipd.it)
Luisa Mancuso (luisa_mancuso@yahoo.it)
Luca Sigalotti (lsigalotti@cro.it)
Barbara Bauce (barbara.bauce@unipd.it)
Gianluca Occhi (gianluca.occhi@unipd.it)
Cristina Basso (cristina.basso@unipd.it)
Gerolamo Lanfranchi (lanfra@cribi.unipd.it)
Jeffrey A Towbin (jtowbin@bcm.tmc.edu)
Gaetano Thiene (gaetano.thiene@unipd.it)
Gian Antonio Danieli (danieli@bio.unipd.it)
Alessandra Rampazzo (alessandra.rampazzo@unipd.it)

Version: 4 Date: 26 October 2007

Author's response to reviews:

MS: 1584464041152603
Missense mutations in Desmocollin-2 N-terminus, associated with arrhythmogenic right ventricular cardiomyopathy, affect intracellular localization of desmocollin-2 in vitro

Giorgia Beffagna, Marzia De Bortoli, Andrea Nava, Michela Salamon, Alessandra Lorenzon, Manuela Zaccurlo, Luisa Mancuso, Luca Sigalotti, Barbara Bauce, Gianluca Occhi, Cristina Basso, Gerolamo Lanfranchi, Jeffrey A Towbin, Gaetano Thiene, Gian Antonio Danieli and Alessandra Rampazzo

TO: The BioMed Central Editorial Production Team
Tel: +44 (0)20 7631 9921
Facsimile: +44 (0)20 7631 9923
e-mail: corepreaccept@biomedcentral.com
Web: http://www.biomedcentral.com/
Padova, October 26, 2007
Please find enclosed the revised version of the above-mentioned paper. Enclosed you will find point-by-point description of the changes made. Looking forward to hearing from you,

Yours sincerely,

Alessandra Rampazzo

Point-by-point description of the changes made.

Materials & Methods - Please rename this section 'Methods'.
This section has been renamed 'Methods'.

Authors' contributions - Please indicate that all authors read and approved the final manuscript.
In the section 'Authors' contributions', the following sentence has been added: All authors read and approved the final manuscript.

Please de-italisise the acknowledgments.
The acknowledgments have been de-italisised.

References - The reference list should contain all authors' names, if there are 30 or below. The term 'et al.' should not be used (ref 3, 12, 15, 17, 24, 26-29).
References have been modified as requested.

We notice table 1 is an embedded figure. We cannot support this. Please create the table from cells as you have done for table 2 and remove the embedded image.
The embedded image has been removed and a novel table has been created.

Moreover, the designation 1034T>C has been incorporated in the legend to Figure 1 and in Table 2, as suggested by Reviewer 1.
Some spelling mistakes have been corrected.